Author's response to reviews

Title: Access to and utilisation of GP services among Burmese migrants in London: a cross-sectional descriptive study

Authors:

Nyein Chan Aung (Nyein.Chan.Aung.Dr@gmail.com)
Bernd Rechel (Bernd.Rechel@lshtm.ac.uk)
Peter Odermatt (peter.odermatt@unibas.ch)

Version: 6 Date: 30 September 2010

Author's response to reviews:

Major revisions (we require the author to make these changes)

Additional files: all additional files must be mentioned in the text in numerical order, or removed from the system. We notice that none of the additional files are cited in the main text.

==> OK, done.

Minor revisions (we can make these changes for you, although it will speed up publication of your manuscript if you do them while making the major changes above)

Tables: please remove all table placement text.

==> Done!

Figures: It is important for the final layout of the manuscript that the figures are cropped as closely as possible to minimise white space around the image. Our online figure guide contains full details for preparing files for submission and can be viewed here: http://www.biomedcentral.com/info/ifora/figures

==> we cropped the figures (now with tiny margins)

Thank you